Updates on pharmacological treatment for Alzheimer's disease

Philip W. Tipton

Department of Neurology, Mayo Clinic Florida, Jacksonville, Florida, United States

ABSTRACT

Alzheimer's disease (AD) is the most common cause of dementia, and its rising prevalence is constantly increasing the global health burden. There are currently no curative therapies for AD, and current treatment options provide only modest clinical benefit. Despite numerous clinical trials, there have been no major additions to the AD treatment armamentarium this century. The prevailing pathomechanistic hypothesis for AD begins with abnormal accumulation of amyloid β (A β) leading to plaque development, and disease-modifying candidate therapies have largely aimed to disrupt this process. Numerous clinical trials of monoclonal antibodies directed at various stages of A β plaque development have yielded mostly negative results; however, recent results suggest that a breakthrough may be on the horizon. The past two years have yielded positive results for three monoclonal antibodies (aducanumab, lecanemab, and donanemab) although questions remain regarding their clinical effectiveness. Additional clarity is needed to determine whether the clinical benefits are great enough to offset the treatment risks and the resource implications for healthcare systems. This review provides a foundational context and update on recent disease-modifying therapies for AD that have reached Phase III clinical trials. Up-to-date information on these therapies will help clinicians better inform their clinical decision-making and the counselling they can offer patients and their carers.

Keywords: dementia, Alzheimer's disease, treatment, amyloid, clinical trials

(Neurol Neurochir Pol 2024; 58 (2): 150-160)

Introduction

Alzheimer's disease (AD) is the most common neurodegenerative disease, with a global prevalence that continues to rise. There are currently 6.1 million individuals with AD in the United States (US) and this is expected to reach 13.8 million by 2060 [1]. Like most neurodegenerative diseases, advancing age is the greatest risk factor in AD and is thought to be the primary driving force behind rising disease prevalence. In 2023, individuals who are 85 years or older account for 33% of all people with AD but this is expected to have risen to 48% by 2060 [1]. The global incidence of AD and other dementias increased by nearly 150% between 1990 and 2019 [2]. These rising figures paint a grim reality for patients, carers, healthcare systems, and the global economy.

Despite these trends, there have been no significant additions to the AD treatment armamentarium since donepezil and memantine were approved by the US Food and Drug Administration (FDA) in 1996 and 2003, respectively. Unfortunately, these medications typically provide only modest benefits to patients. In 2016, AD became the 6th most burdensome disease/injury in the US according to the Global Burden of Disease classification system, and it was the seventh-leading cause of death in the US in 2021 [3, 4]. The need for treatments that can prevent or significantly alter the trajectory of disease has never been greater. A treatment that could delay the onset of AD-related symptoms by five years would result in 41% lower prevalence and 40% lower cost of AD in 2050 [5].

The distinct histopathological features initially witnessed by Dr. Alois Alzheimer include plaques composed primarily of extracellular beta amyloid (A β), and neurofibrillary tangles, which consist of intraneuronal phosphorylated tau. Neurological symptoms correlate with the distribution of neurofibrillary tangles rather than A β plaques [6]. We now know that A β accumulation and plaque formation precede

Address for correspondence: Philip W. Tipton, M.D., Mayo Clinic, 4500 San Pablo Rd, Jacksonville, FL 32224, USA; e-mail: tipton.philip@mayo.edu Received: 30.06.2023 Accepted: 25.07.2023 Early publication date: 22.08.2023

This article is available in open access under Creative Common Attribution-Non-Commercial-No Derivatives 4.0 International (CC BY-NC-ND 4.0) license, allowing to download articles and share them with others as long as they credit the authors and the publisher, but without permission to change them in any way or use them commercially.





Figure 1. Monoclonal antibody affinity for amyloid β plaque development stages. Illustration of staged amyloid β plaque development and primary affinities for anti-amyloid monoclonal antibodies with Phase III clinical trial results. Bar segment colour intensity approximately correlates with binding affinity of antibody. Image of amyloid plaque visualised with thioflavin staining

tau tangle formation, which precedes neurodegeneration [7]. Some have questioned the validity of this amyloid hypothesis, yet these pathological hallmarks remain the primary focus of disease-modifying treatment trials [8, 9]. Unique treatment side effects have subsequently emerged, including amyloid-related imaging abnormalities (ARIA), which occur more commonly among individuals who carrier the APOE ɛ4 allele. These individuals are known to have an increased risk of AD [10].

The overall drug development pipeline for AD changes substantially each year, but the group of treatments in Phase III clinical trials remains relatively stable due to long trial durations. Nevertheless, there have been major updates in the status of many drugs currently in large scale trials. After decades of failed therapy trials, it has suddenly become a challenge to remain informed about the status of various treatments as they move through the research and development pipeline. This review aims to provide an update on the current status and trajectory of the disease-modifying treatment landscape for AD.

Anti-amyloid therapies

Deposition of the A β protein is the earliest histopathological hallmark of AD, and begins in the neocortex before propagating, in a predictable fashion, to deeper brain structures. Thal et al. [11] characterised the A β propagation pattern to involve the CA1 region of the hippocampus and entorhinal cortex (Thal phase 2) before progressing through the thalamus, basal ganglia, and other structures ultimately including the cerebellum (Thal phase 5). Because this deposition pattern is well underway before tau tangle accumulation or neurodegeneration, therapies targetting A β have dominated the clinical trial space for nearly two decades. Since the earliest Phase III clinical trials of bapineuzumab, a number of monoclonal antibodies have been developed to target various regions of the A β protein with varying degrees of specificity for different stages of A β plaque formation (Fig. 1). Unfortunately, every drug failed in its Phase III clinical trial, until 2021.

Aducanumab

Aducanumab (Aduhelm; Biogen, Inc.) is a humanised IgG1 monoclonal antibody that targets aggregated forms of A β and is administered as a monthly intravenous (i.v.) infusion. In 2012, a double-blind, placebo-controlled Phase Ib randomised trial (PRIME) of patients with prodromal or mild AD began. Interim analysis of 165 patients who received either the drug or a placebo for one year showed dose-dependent target engagement, based upon significant Aß plaque reduction on florbetapir PET imaging [12]. While this study was not powered to detect clinical change, there was evidence supporting a dose-dependent slowing of clinical progression based on differences in the change from baseline on the clinical dementia rating scale sum of boxes (CDR-SB) and the Mini Mental State Examination (MMSE) [13]. The most common adverse effect was ARIA, which occurred more often among participants receiving higher drug doses and those who were APOE £4 allele carriers. ARIA occurred with both vasogenic oedema (ARIA-E) and haemorrhagic (ARIA-H) manifestations, but was symptomatic in a minority of patients.

In 2015, two double-blind, placebo-controlled, parallel-group Phase III randomised clinical trials (EMERGE and ENGAGE) began for individuals with mild cognitive impairment (MCI) or mild dementia due to AD confirmed with amyloid PET. Participants were randomised in a 1:1:1 ratio to receive low dose aducanumab (3 mg/kg), or high dose aducanumab (6 mg/kg), or a placebo [14]. Based on the previous data indicating a greater risk of ARIA among APOE ϵ 4 allele carriers, only non-carriers of the ϵ 4 allele were eligible to receive 10 mg/kg doses. After multiple failed trials indicated that ARIA-E was most often asymptomatic and reversible, the study protocols for EMERGE and ENGAGE were modified to permit APOE ϵ 4 allele carriers to receive the 10 mg/kg dose [15–17].

Table 1. Ongoing or planned phase III/IV	/ trials of disease-modifying therapies for AD
--	--

Drug	Trial	Study population	Intervention	Primary endpoint	Expected com- pletion date
Aducanumab	NCT05310071 (ENVI- SION)	MCI or mild dementia due to AD	i.v. infusion of aducanu- mab q4 wks	Change from baseline in CDR-SB	10/2026
	NCT05108922 (TRAIL- BLAZER-ALZ 4)	Early symptomatic AD (CDR-GS 0.5 or 1) & MMSE 20–30)	i.v. infusion of aduca- numab or donanemab q4 wks	% of Participants who reach complete amyloid plaque clearance on PET	7/2024
	NCT04241068 Open-Label Extension	MCI or mild dementia due to AD	i.v. infusion of aducanu- mab q4 wks	AE, serious AE, ARIA-E, ARIA-H	2/2025
Donanemab	NCT05108922 (TRAIL- BLAZER-ALZ 4)	Early symptomatic AD (CDR-GS 0.5 or 1) & MMSE 20–30)	i.v. aducanumab or donanemab q4 wks	% of Participants who reach complete amyloid plaque clearance on PET.	7/2024
	NCT05026866 (TRAILBLAZER-ALZ 3)	Cognitively normal AD	i.v. donanemab or placebo q4 wks	Change from baseline in CDR-GS.	11/2027
	NCT05508789 (TRAILBLAZER-ALZ 5) *Europe & Asia only	Early symptomatic AD (MMSE 20–28)	i.v. donanemab or placebo q4 wks	Change from baseline on iADRS	6/2027
	NCT05738486 (TRAILBLAZER-ALZ 6)	Early symptomatic AD (MMSE 20–28)	i.v. donanemab or placebo at varying intervals	ARIA-E	5/2025
Lecanemab	NCT04468659 (AHEAD 3-45)	Cognitively normal AD	i.v. lecanemab or place- bo q2 wks	A45: Change from baseline in PACC5 A3: Change from baseline in amyloid PET	10/2027
	NCT05269394 (DIAN-TU)	Cognitively normal or MCI with known deter- ministic AD mutation	i.v. lecanemab & E2814	Change in tau PET	10/2027
	NCT01760005 (DIAN-TU-001)	Cognitively normal or MCI with known deter- ministic AD mutation	i.v. lecanemab & E2814	Change from baseline in DIAN-MCE	10/2027
E2814	See lecanemab for NCT05269394 & NCT01760005				
Remternetug	NCT05463731 (TRAILRUNNER-ALZ 1)	Early symptomatic AD (MMSE 20–28)	i.v. or SC remternetug or placebo	% of participants who reach complete amyloid plaque clearance on PET	10/2026
Tertomotide	NCT05303701	Severe AD (NINCDS- -ADRDA criteria and Korean MMSE ≤ 19)	SC tertomotide or placebo q4 wks then q2 wks	Change from baseline in SIB and CDR-SB	4/2026
Semaglutide	NCT04777396 (EVOKE)	MCI or mild dementia due to AD (CDR-GS 0.5 or 1.0 & MMSE ≥ 22)	Oral semaglutide or placebo daily	Change from baseline in CDR-SB	10/2026
	NCT04777409 (EVOKE Plus)	MCI or mild dementia due to AD (CDR-GS 0.5 or 1.0 & MMSE ≥ 22)	Oral semaglutide or placebo daily	Change from baseline in CDR-SB	10/2026
	NCT05891496	MCI or mild dementia due to AD (CDR-GS 0.5 or 1.0)	SC semaglutide or placebo weekly	Change in gene expression with scRNAseq (CSF and blood)	6/2025

AE — adverse events; ARIA-E — amyloid related imaging abnormalities oedema subtype; ARIA-H — amyloid related imaging abnormalities haemorrhagic subtype; CDR-SB — Clinical Dementia Rating Scale--Sum of Boxes; CDR-Go — CDR-Global Score; CSF — cerebrospinal fluid; DIAN-MCE — Dominantly Inherited Alzheimer's Network-Multivariate Cognitive Endpoint; iADRS — Integated Alzheimer's Disease Rating Scale; i.v. — intravenous; MCI — mild cognitive impairment; NINCDS-ADRDA — National Institute of Neurological and Communicative Diseases and Stroke/Alzheimer's Disease Rascotation; PACC5 — Preclinical Alzheimer's Cognitive Composite 5; SC — subcutaneous; SIB — Severe Impairment Battery. According to clinicaltrials.gov as of 25 July, 2023

In 2019, an interim analysis predicted that neither trial would meet its primary endpoint of a slowed rate of decline on the CDR-SB. However, data acquired during the time between this announcement and the actual trial stop date indicated that EMERGE met its primary endpoint because participants receiving the highest drug dose (10 mg/kg) showed a significant reduction in decline on the CDR-SB [18]. These participants also met the secondary endpoints of slowed decline on the MMSE, Alzheimer's Disease Assessment Scale-Cognitive subscale (ADAS-Cog), and Alzheimer's Disease Cooperative Study Activities of Daily Living for Mild Cognitive Impairment (ADCS-ADL-MCI). Based on this, and an exploratory analysis suggesting that ENGAGE participants who received \geq 10 doses of the 10 mg/kg dose declined more slowly, Biogen applied for approvals from the US FDA, the European Union, and Japan. Despite the recommendation of an FDA advisory panel to deny approval, based on the drug's inability to meet the primary endpoint in both Phase III clinical trials, the FDA approved aducanumab in the summer of 2021. The FDA also required a Phase IV confirmatory trial that would eventually be titled ENVISION (Tab. 1).

As the first disease-modifying therapy approved for AD, aducanumab was initially priced at \$56,000 per annum. Biogen Inc. lowered the drug price to \$28,200 partially out of growing concern for the economic implications of prescribing such an expensive medication in a large percentage of the population, most of whom are enrolled in Medicare (in the US). Based on this information, and the need for additional neuroimaging surveillance, the estimated annualised Medicare costs were \$7 billion if prescribing strictly adhered to the clinical trial inclusion criteria [19]. This figure rose steeply to \$37.4 billion when taking into account off-label prescribing. Ultimately, the US Centres for Medicare and Medicaid Services (CMS) restricted coverage to individuals in clinical trials. In 2022, Biogen Inc. withdrew its Marketing Authorisation Application to the European Medicines Agency. Aducanumab remains available in the US only for patients who are able to pay directly out of their own pockets, given the lack of insurance coverage.

In addition to ENVISION, which will conclude in 2026, ongoing studies of aducanumab include an open label extension of EMERGE/ENGAGE, a head-to-head comparison with donanemab, and a phase Ib open-label study to evaluate the safety and feasibility of opening the blood-brain barrier in conjunction with aducanumab therapy (Tab. 1) [20].

Donanemab

Donanemab (Eli Lilly & Co.) is a humanised IgG1 monoclonal antibody that directly targets Aß plaques. Results from a Phase I study of participants with MCI or mild dementia due to AD demonstrated safety over a 12-week follow-up period [21]. Participants received a single i.v. infusion and only the highest dose (10 mg/kg) yielded amyloid reduction at study completion. A subsequent randomised placebo-controlled Phase II study (TRAILBLAZER-ALZ) included 257 participants with MCI or mild dementia due to AD (confirmed with amyloid and tau PET positivity). Participants in the treatment arm received 700 mg for the first three monthly infusions before doubling the dose [22]. At the completion of a 76-week study period, the treatment group showed a statistically significant 32% reduction in the rate of decline on the Integrated Alzheimer's Disease Rating Scale (iADRS). There were no differences among secondary outcomes including CDR-SB, ADAS-Cog₁₃, Alzheimer's Disease Cooperative Study-Instrumental Activities of Daily Living Inventory (ADCS-iADL), and MMSE, nor changes in amyloid or tau brain PET signals.

Post-hoc analyses of TRAILBLAZER-ALZ showed that the rate of amyloid reduction at 24 weeks correlated with the amount of baseline amyloid, and modelling suggested that amyloid would not reaccumulate to the AD threshold for nearly four years after drug discontinuation [23]. Among participants with complete amyloid clearance, tau accumulated more slowly. A separate analysis of study participants found that APOE £4 allele carriers were four times more likely than non-carriers to have ARIA-E by 24 weeks [24]. Participants receiving donanemab showed significant reductions in plasma pTau217 and glial fibrillary acidic protein, although neurofilament light chain (NfL) was not significantly changed [25]. An open-label extension including a one-month validation of home assessments is expected to be complete by early 2024 [26].

TRAILBLAZER-ALZ2 is a Phase III, double-blind, placebo-controlled study of 1,736 participants with prodromal AD and mild dementia due to AD (confirmed with 18F flortaucipir and florbetapir PET scans). The study met its primary endpoint of a slowed rate of change from baseline on the iADRS following 76 weeks of treatment. After one year, 47% of participants receiving donanemab showed no decline on the CDS-SB compared to only 29% of participants receiving a placebo [27]. Among all participants receiving donanemab, there was a 22.3% slowing compared to the placebo. Study investigators also stratified participants based upon tau burden and demonstrated that those with low/medium tau showed an even greater benefit from therapy (35.1% slowing). These important findings suggest that there are subgroups of patients that will respond more favourably to this therapy and may have implications for future therapy decisions.

In 2021, Eli Lilly & Co. began TRAILBLAZER-ALZ3, which is a double-blind, randomised, placebo-controlled trial of cognitively normal participants at risk for developing symptomatic AD based upon positive plasma p-tau-217. The primary outcome measure is the time to clinical progression measured by the CDR global score (CDR-GS) after 3.5 years of follow-up. A unique aspect of this trial is the implementation of the modified Telephone Interview for Cognitive Status (TICS-M) (Tab. 1) [28]. TRAILBLAZER-ALZ4 is a head-to-head comparison of donanemab versus aducanumab with respect to amyloid plaque clearance in participants with early symptomatic AD, defined as a CDR-GS of 0.5-1 and MMSE of 20-30. The estimated study completion is mid-2024 (Tab. 1) [29]. TRAILBLAZER-ALZ5 has an identical trial design to that of TRAILBLAZER-ALZ2, but will include only study sites in Europe (including Poland) and Asia (Tab. 1) [30].

Gantenerumab

 $Gantenerumab\,(F.\,Hoffmann-La\,Roche)\,is\,an\,IgG1\,antibody\\ directed against A\beta\,fibrils that is administered by subcutane$ $ous\,(SC) injection. The SCarlet RoAD trial was a randomised,$

double-blind, placebo-controlled phase II study assessing the safety and efficacy of gantenerumab administered in low (105 mg) or high (225 mg) doses every four weeks to participants with prodromal AD for two years [31, 32]. This trial was also the first to include amyloid biomarker confirmation as an inclusion criterion [16]. Two years into the study, results from a Phase I study of gantenerumab supported A β clearance [33]. Subsequently, Roche converted SCarlet RoAD into a Phase III study and began a fresh Phase III study (Marguerite RoAD) featuring only participants with mild AD dementia. Two years later, a futility analysis of SCarlet RoAD indicated that the primary outcome measure was unlikely to be met. Study investigators attributed this negative result to low drug doses, which were chosen to minimise the risk of ARIA. However, Phase III trials of bapineuzumab indicated that ARIA-E tended to be asymptomatic and was more likely among APOE E4 allele carriers [34, 35]. As a result of the SCarlet RoAD futility analysis, dosing was interrupted, but participants continued to be followed up for safety and efficacy. Recruitment for Marguerite RoAD was stopped, but dosing continued.

In 2013, modelling estimations of the AD neuroimaging initiative (ADNI) database clustered AD patients into fast-progressing and slow-progressing subpopulations [36]. By applying this model to the SCarlet RoAD cohort, the trial investigators concluded that fast progressors demonstrated a dose-dependent slowing of decline in ADAS-Cog₁₃, Cambridge Neuropsychological Test Automated Battery (CANTAB), and MMSE. Taken together with significant reductions in levels of CSF t-tau and p-tau as well as positive Phase Ib results for high-dose aducanumab, it was speculated that gantenerumab was dosed too low [12, 16]. Subsequently, SCarlet RoAD and Marguerite RoAD were both converted into open label extensions to test the safety of higher doses up to 1,200 mg [37, 38]. An amyloid PET sub-study showed that 51% and 80% of participants were below the amyloid positivity threshold at two and three years, respectively [39, 40].

In 2018, the GRADUATE I and GRADUATE II parallel, randomised, double-blind, placebo-controlled trials began across 30 different countries. These twin trials assessed the efficacy and safety of subcutaneous (SC) injections of gantenerumab initially dosed at 120 mg every four weeks and increased to 510 mg every two weeks by the 9th month regardless of APOE allele status [31]. In November 2022, Roche announced that both GRADUATE studies did not meet their primary endpoint of a slowed rate of decline on the CDR-SB [41]. The following month, Roche presented unpublished data at the Clinical Trials on Alzheimer's disease conference that indicated that gantenerumab cleared fewer A β plaques than anticipated. The company then discontinued SKYLINE, a Phase III secondary prevention trial, that had begun in early 2022 [42].

In 2012, the Dominantly Inherited Alzheimer's Network Trials Unit (DIAN-TU) began a randomised, placebo-controlled, Phase II/III trial (DIAN-TU-001) testing

gantenerumab and solanezumab in asymptomatic (CDR 0) and mildly symptomatic (CDR 0.5) carriers of mutations in APP, PSEN1, or PSEN2 genes [43]. The study failed to reach its primary endpoint of showing a difference compared to a placebo on the DIAN Multivariate Cognitive End Point (DIAN-MCE), which is a composite metric of cognition [44]. While solanezumab showed no significant improvements on disease biomarkers, gantenerumab reduced amyloid plaques, CSF t-tau, phospho-tau181, and slowed increases in NfL levels [44]. An open-label extension continued with gantenerumab. Initially, the DIAN-TU investigators were planning to launch a primary prevention trial (DIAN-TU-002) of gantenerumab with an anti-tau agent; however, this was halted after results of DIAN-TU-001 and the GRADUATE studies were released, and Roche discontinued development of the drug in its current form. Roche is now developing a new version (trontinemab) to enhance blood-brain barrier penetration.

Solanezumab

Solanezumab (Eli Lilly & Co.) is a humanised monoclonal IgG1 antibody directed against soluble Aß delivered by monthly i.v. infusions. Phase III randomised, placebo-controlled clinical trials of solanezumab, including EXPEDITION-1 and EXPEDITION-2, included more than 2,000 participants with mild-to-moderate AD. Primary outcome measures of ADAS-Cog₁₁ and ADCS-ADL showed no significant difference compared to a placebo [45]. A subsequent double-blind, placebo-controlled phase III trial (EXPEDITION-3) limited enrollment to participants with mild AD, but also failed to show a significant slowing of cognitive decline measured on the ADAS-Cog14 [15]. Despite these disappointing results, a positive finding was the rare occurrence of ARIA that contributed to other study investigators deciding to increase dosages in other anti-amyloid monoclonal antibody trials, i.e. EMERGE and ENGAGE. Solanezumab also failed to slow the rate of cognitive decline among participants in DIAN-TU-001 as previously discussed (see gantenerumab section) [44].

In 2014, Lilly began the A4 (Anti-Amyloid Treatment in Asymptomatic AD) Phase III clinical trial with a primary outcome measure of change in the Preclinical Alzheimer's Cognitive Composite (PACC) score. In March 2023, Lilly announced that despite quadrupling the initial dosage, an adjustment made in response to EXPEDITION-3 showing low ARIA rates, four years of treatment had not slowed cognitive decline, nor reduced the risk of progression to symptomatic AD, nor resulted in the clearance of A β plaque. There are currently no active or planned studies featuring solanezumab (Tab. 1).

Lecanemab

Lecanemab (Eisai Co., Ltd.) is a humanised IgG1 monoclonal antibody that binds soluble A β protofibrils and is administered by monthly i.v. infusions. Initial human Phase I studies of lecanemab, at that time named BAN2401, found dose tolerability up to 10 mg/kg every two weeks among

participants with mild-to-moderate AD [46]. A dose exploration Phase II study assessed participants with early-stage AD (NIA-AA criteria), MCI due to AD, or probable AD (NIA-AA criteria) with amyloid PET confirmation. Despite a Bayesian adaptive design, full enrollment of 854 participants was required because this study failed to meet either its primary endpoint or futility conditions at any of the 17 interim analyses [47]. The primary endpoint required 80% probability of $\ge 25\%$ reduction in clinical decline at 12 months on the Alzheimer's Disease Composite Score of cognitive scores (ADCOMS) [48]. While the primary endpoint was not met, Bayesian and frequentist analyses of secondary endpoints at 18 months indicated that 10 mg/kg biweekly lecanemab decreased signal on amyloid PET and decreased the rate of decline on clinical measures (ADCOMS, ADAS-Cog14, CDR-SB) [49]. A 24-month open-label extension of 10 mg/kg biweekly dosing began after a drug-free gap (average of 24 months) during which AD biomarkers, particularly plasma A $\beta/40$ ratio and p-tau181 levels, began to return to pre-treatment levels while clinical differences from the placebo persisted. Resuming lecanemab resulted in a significantly reduced amyloid plaque and rate of cognitive decline [50].

In 2019, Eisai began a double-blind, Phase III clinical trial (Clarity AD) that enrolled 1,795 participants with MCI or mild dementia due to AD (CSF or PET confirmed), who were randomised to receive i.v. lecanemab 10 mg/kg or a placebo every two weeks. In January 2023, published results showed that lecanemab met its primary endpoint of slowing the rate of decline on CDR-SB at 18 months compared to the placebo. The difference in adjusted least-squares mean change from baseline was -0.45 in favour of lecanemab [51]. Secondary endpoints indicated greater reductions in brain amyloid burden and drug-favouring differences for the ADAS-Cog₁₄, ADCOMS, and ADCS-MCI-ADL score. The most common adverse event was infusion-related reactions, which occurred in 26.4% of participants. ARIA-E occurred in 12.6% of participants. More than half of these participants were homozygous for the APOE £4 allele. Among all APOE £4 homozygotes, nearly one in three experienced ARIA-E, although less than 10% were symptomatic. ARIA-H occurred in 17.3% of patients receiving lecanemab compared to 9.0% receiving a placebo; however, symptomatic cases only occurred in 0.7% and 0.2% of participants receiving lecanemab or a placebo, respectively.

While Clarity AD is the first unequivocally positive Phase III clinical trial of an anti-amyloid monoclonal antibody, the question remains as to whether the statistically significant findings are clinically significant. A retrospective analysis of the National Alzheimer's Coordinating Centre Uniform Data Set indicated that a minimal clinically important difference (MCID) in CDR-SB is 0.98 and 1.63 for MCI and mild dementia due to AD, respectively [52]. A 3-year multicentre analysis of participants in the Alzheimer's Disease Cooperative Study indicated that the 12-month MCID for CDR-SB in MCI is 1.0–2.5 [53]. Lecanemab's annual difference was only 0.45 and while this may translate into clinical significance over several years, long-term follow-up data is needed for confirmation. Additional concerns have been raised regarding study methodology and interpretation of results, such as the effect of functional unblinding due to adverse events like ARIA [54]. The US FDA approved lecanemab through the Accelerated Approval Programme and granted full approval in July 2023. The European Medicines Agency is currently reviewing a regulatory application for lecanemab [55]. While the US Veterans Health Administration has agreed to pay the \$26,500 drug cost, the US Centres for Medicare & Medicaid Services (CMS) is expected to make coverage decisions after the FDA's final approval verdict. It is important to consider ancillary costs associated with providing lecanemab. These should include MRI surveillance for ARIA, procedural costs associated with administering i.v. infusions, and potential costs associated with managing infusion-related reactions and therapy-related adverse events, e.g. ARIA [56]. Fortunately, ARIA is typically asymptomatic; however, clinicians should carefully consider the risks and potential benefits when determining whether anti-amyloid therapy should be continued and/or supportive therapies, e.g. steroids, should be administered [56, 57]. An estimated 5.4 million Europeans will be eligible for lecanemab and the associated treatment costs will probably exceed 133 billion EUR per year if European drug prices are similar to those of the US [58].

Ongoing trials of lecanemab include the Clarity AD open-label extension, which has reported three deaths due to haemorrhage. Two of these individuals received blood thinners and the third had preexisting amyloid angiopathy [59]. In 2020, the Alzheimer's Clinical Trial Consortium (ACTC) began AHEAD 3-45, which is a 4-year Phase III study of lecanemab in cognitively normal participants with elevated brain amyloid [60]. AHEAD 3-45 consists of two sub-studies. The A3 substudy includes a target enrollment of 400 participants whose amyloid PET is below the positivity threshold. Participants are randomised to receive monthly i.v. infusions of a placebo or lecanemab 5 mg/kg during weeks 0 to 4, and 10 mg/kg thereafter. The primary outcome measure is a change in amyloid PET after four years of treatment. The A45 sub-study plans to enroll 1,400 participants with positive amyloid PET scans. A45 participants are randomised to receive a biweekly placebo or lecanemab 5 mg/kg during weeks 0 to 6 and 10 mg/kg from weeks 8 to 94 followed by monthly 10 mg/kg until study completion. The primary outcome measure is a change from baseline of the PACC5 after four years of treatment.

Remternetug

Remternetug (Eli Lilly & Co.) is a monoclonal antibody that targets a pyroglutamated form of A β aggregated in amyloid plaques like donanemab, with the key difference that remternetug can be administered via SC injection or i.v. infusion. In 2020, a Phase I study began enrolling participants with clinically diagnosed MCI due to AD, dementia due to AD, and healthy volunteers. Participants would be randomised to receive remternetug (i.v. or SC), in escalating doses, or a placebo. The primary outcome measure is the occurrence of one or more serious adverse events after 61 weeks of drug/placebo. Secondary outcome measures include pharmacokinetic properties (maximum concentration and area under the concentration vs. time curve) and a change from baseline in florbetapir PET signal [61]. In 2023, Lilly presented interim data at the annual AD/PD Conference showing dose-dependent plaque reduction [62]. They also showed that every study participant receiving the highest dose fell below the amyloid PET positivity within three months. All ARIA occurrences occurred in APOEɛ4 carriers. This study is expected to run until the middle of 2024.

In 2022, Lilly launched a Phase III, randomised, double--blind, placebo-controlled study (TRAILRUNNER-ALZ1) to evaluate remternetug in participants with early symptomatic AD. Participants had to have gradual and progressive cognitive decline for at least six months prior to screening, score within the range of 20-28 on the MMSE, and have a p-tau and amyloid PET profile consistent with AD [63]. This study will enroll 600 participants, who will receive remternetug or a placebo administered in either SC injection or i.v. infusion for one year followed by a one year cross-over extension period. The study will enroll an additional 640 participants into an addendum safety cohort. These participants will not participate in the extension period. The study's primary outcome measure is a difference in the percentage of participants who reach amyloid plaque clearance on amyloid PET. Secondary outcomes include other amyloid PET-related metrics, observed serum trough concentration of remternetug, and anti-drug antibody detection. TRAILRUNNER-ALZ1 has an estimated completion date of October 2026.

Other pharmacological therapies

E2814

E2814 (Eisai Co., Ltd.) is a humanised monoclonal -antibody directed against an epitope within the microtubule--binding domain of the of 4R and 3R repeat tau isoforms administered by i.v. infusion [64]. A Phase I trial (NCT04231513) to test safety and tolerability in healthy volunteers was completed in 2020 with no evidence of drug-related adverse events [65]. Using liquid chromatography-mass spectrometry, the investigators identified a dose-related increase in antibody-associated tau that persisted for at least one month indicating target engagement [66]. In 2021, E2814 was chosen to be used with lecanemab in the DIAN-TU-001 prevention trial [67] to assess symptomatic (MCI or mild dementia) and asymptomatic populations. The symptomatic population are receiving open-label biweekly lecanemab infusions for 24 weeks after which they will be randomised to receive lecanemab plus either intravenous E2814 or a placebo. The asymptomatic population are being randomised to begin the study with either E2814 or a placebo. After one year, open label lecanemab will be added to all participants. The primary outcome measure is a change in tau PET signal from week 24 to week 104 and to week 208 for the symptomatic population. Secondary outcome measures include changes in CDR-SB, a cognitive composite, amyloid PET, and CSF NfL for symptomatic participants. Secondary outcome measures for the asymptomatic population include a change in CSF p-tau217/t-tau ratio and CSF NfL. Study results are expected in 2027. These results are being eagerly anticipated as it has been proposed that combination therapies, e.g. anti-amyloid plus anti-tau, may prove more effective [68].

Tertomotide

Tertomotide (GemVax & KAEL Bio), also known as GV1001 or RIAVAXTM, was initially developed as a peptide vaccine that targets telomerase reverse transcriptase (TERT), which is highly expressed by many cancers. Although a Phase III trial to treat pancreatic cancer was negative, the drug has been repurposed to treat AD based on neuroprotective effects against $A\beta_{25-35}$ oligomer in rodent neural stem cells [69, 70]. In rodents treated with hydrogen peroxide, tertomotide reduced free radical levels and restored expression of survival-related proteins while reducing expression of those proteins associated with cellular death [71].

In 2019, a phase II, double-blind, parallel-group, placebo-controlled, 6-month randomised clinical trial of participants with moderate-to-severe AD began in Korea. Ninety-six participants were randomised in a 1:1:1 fashion to receive four weekly SC injections of low-dose tertomotide, or high-dose tertomotide, or a placebo followed by 10 injections every two weeks. The study met its primary endpoint of a change in the Severe Impairment Battery (SIB) from baseline to 24 weeks [72]. While secondary endpoints, including change on the neuropsychiatric inventory (NPI), ADCS-ADL, and CDR-SB, trended similarly to the SIB, only NPI changes reached statistical significance. The treatment and placebo groups did not differ with respect to the incidence of adverse events.

Currently, a multicentre, randomised, double-blind, placebo-controlled, parallel-design Phase II study in the US is following participants with mild-to-moderate AD randomised to low/high dose tertomotide or a placebo. The primary outcome measure is a change from baseline in ADAS-Cog₁₁ after one year [73]. This study is expected to complete by September 2024. In early 2022, a Korean Phase III study began to assess participants with moderate-to-severe AD with an MMSE score \leq 19 [74]. Study participants were randomised to receive four weekly SC injections of low/high dose tertomotide or a placebo, then 10 injections administered every other week followed by an open extension phase of high dose tertomotide for 49 weeks. The primary outcome measure is a change from baseline in the SIB and CDR-SB. This study is set to complete in early 2026.

Semaglutide

Semaglutide (Novo Nodisk A/S) is a synthetic, long-acting analogue of glucagon-like peptide-1 that is currently used to treat diabetes under the brand names OzempicTM and RybelsusTM. Based on studies showing that GLP-1 crosses the blood-brain barrier and may improve learning and memory in rodents, theories have emerged regarding a neuroprotective effect in neurodegenerative diseases, e.g. AD and Parkinson's disease (PD) [75–78].

In 2021, a randomised, double-blind, placebo-controlled Phase III study (EVOKE) of semaglutide began for participants with early AD (MCI or mild dementia with AD biomarker confirmation) [79]. Study participants were randomised to receive either once daily oral semaglutide (14 mg) or a placebo for up to 173 weeks. The primary outcome measure is a change in the CDR-SB after two years of treatment. Secondary outcome measures include ADCS-ADL-MCI, time to dementia conversion, ADCOMS, MMSE, NPI, c-reactive protein level changes, and others. A nearly identical Phase III study (EVOKE Plus) will include different cognitive measures, e.g. Montreal Cognitive Assessment and the ADAS-Cog13 [80]. Both EVOKE and EVOKE plus will have a one year extension phase, and are expected to complete in late 2026.

Conclusion

After years that have witnessed many failed Phase III clinical trials, there now appear to be encouraging signs given recent positive results for two anti-amyloid monoclonal antibodies. While questions remain regarding the clinical significance, associated risks, and burden to healthcare systems, many patients are understandably eager to have options at their disposal. The impact that these therapies will have on the AD treatment landscape remains to be seen. This review provides a foundation upon which clinicians can build so that they can stay abreast of updates in this ever-changing field.

Article information

Author contributions: PWT: Completed all aspects of manuscript preparation. Funding: None. Acknowledgments: Thank you to Dr. Dennis Dickson for providing the amyloid plaque imaging in Figure 1. Conflict of interest: None. Supplementary material: None.

References

 Rajan KB, Weuve J, Barnes LL, et al. Population estimate of people with clinical Alzheimer's disease and mild cognitive impairment in the United States (2020-2060). Alzheimers Dement. 2021; 17(12): 1966–1975, doi: 10.1002/alz.12362, indexed in Pubmed: 34043283.

- Li X, Feng X, Sun X, et al. Global, regional, and national burden of Alzheimer's disease and other dementias, 1990-2019. Front Aging Neurosci. 2022; 14: 937486, doi: 10.3389/fnagi.2022.937486, indexed in Pubmed: 36299608.
- 2023 Alzheimer's disease facts and figures. Alzheimers Dement. 2023; 19(4): 1598–1695, doi: 10.1002/alz.13016, indexed in Pubmed: 36918389.
- Mokdad AH, Ballestros K, Echko M, et al. US Burden of Disease Collaborators. The State of US Health, 1990-2016: Burden of Diseases, Injuries, and Risk Factors Among US States. JAMA. 2018; 319(14): 1444–1472, doi: 10.1001/jama.2018.0158, indexed in Pubmed: 29634829.
- Zissimopoulos J, Crimmins E, St Clair P. The Value of Delaying Alzheimer's Disease Onset. Forum Health Econ Policy. 2014; 18(1): 25–39, doi: 10.1515/fhep-2014-0013, indexed in Pubmed: 27134606.
- Brier MR, Gordon B, Friedrichsen K, et al. Tau and Aβ imaging, CSF measures, and cognition in Alzheimer's disease. Sci Transl Med. 2016; 8(338): 338ra66, doi: 10.1126/scitranslmed.aaf2362, indexed in Pubmed: 27169802.
- Jack CR, Holtzman DM. Biomarker modeling of Alzheimer's disease. Neuron. 2013; 80(6): 1347–1358, doi: 10.1016/j.neuron.2013.12.003, indexed in Pubmed: 24360540.
- Cummings J, Zhou Y, Lee G, et al. Alzheimer's disease drug development pipeline: 2023. Alzheimers Dement (N Y). 2023; 9(2): e12385, doi: 10.1002/trc2.12385, indexed in Pubmed: 37251912.
- Makin S. The amyloid hypothesis on trial. Nature. 2018; 559(7715): S4–S7, doi: 10.1038/d41586-018-05719-4, indexed in Pubmed: 30046080.
- Tipton PW, Bülbül N, Crook J, et al. Effects of sex and APOE on Parkinson's Disease-related cognitive decline. Neurol Neurochir Pol. 2021; 55(6): 559–566, doi: 10.5603/PJNNS.a2021.0071, indexed in Pubmed: 34642926.
- Thal DR, Rüb U, Orantes M, et al. Phases of A beta-deposition in the human brain and its relevance for the development of AD. Neurology. 2002; 58(12): 1791–1800, doi: 10.1212/wnl.58.12.1791, indexed in Pubmed: 12084879.
- Sevigny J, Chiao P, Bussière T, et al. The antibody aducanumab reduces Aβ plaques in Alzheimer's disease. Nature. 2016; 537(7618): 50–56, doi: 10.1038/nature19323, indexed in Pubmed: 27582220.
- Sokołowska N, Sokołowski R, Oleksy E, et al. Usefulness of the Polish versions of the Montreal Cognitive Assessment 7.2 and the Mini-Mental State Examination as screening instruments for the detection of mild neurocognitive disorder. Neurol Neurochir Pol. 2020; 54(5): 440–448, doi: 10.5603/PJNNS.a2020.0064, indexed in Pubmed: 32808669.
- Salloway S, Chalkias S, Barkhof F, et al. Amyloid-Related Imaging Abnormalities in 2 Phase 3 Studies Evaluating Aducanumab in Patients With Early Alzheimer Disease. JAMA Neurol. 2022; 79(1): 13–21, doi: 10.1001/jamaneurol.2021.4161, indexed in Pubmed: 34807243.
- Honig LS, Vellas B, Woodward M, et al. Trial of Solanezumab for Mild Dementia Due to Alzheimer's Disease. N Engl J Med. 2018; 378(4): 321– 330, doi: 10.1056/NEJMoa1705971, indexed in Pubmed: 29365294.
- Ostrowitzki S, Lasser RA, Dorflinger E, et al. SCarlet RoAD Investigators, SCarlet RoAD Investigators. A phase III randomized trial of gantenerumab in prodromal Alzheimer's disease. Alzheimers Res Ther. 2017; 9(1): 95, doi: 10.1186/s13195-017-0318-y, indexed in Pubmed: 29221491.
- Salloway S, Honigberg LA, Cho W, et al. Amyloid positron emission tomography and cerebrospinal fluid results from a crenezumab anti-amyloid-beta antibody double-blind, placebo-controlled, randomized phase II study in mild-to-moderate Alzheimer's disease (BLAZE).

Alzheimers Res Ther. 2018; 10(1): 96, doi: 10.1186/s13195-018-0424-5, indexed in Pubmed: 30231896.

- Budd Haeberlein S, Aisen PS, Barkhof F, et al. Two Randomized Phase 3 Studies of Aducanumab in Early Alzheimer's Disease. J Prev Alzheimers Dis. 2022; 9(2): 197–210, doi: 10.14283/jpad.2022.30, indexed in Pubmed: 35542991.
- Mafi JN, Leng M, Arbanas JC, et al. Estimated Annual Spending on Aducanumab in the US Medicare Program. JAMA Health Forum. 2022; 3(1): e214495, doi: 10.1001/jamahealthforum.2021.4495, indexed in Pubmed: 35977233.
- Safety and Feasibility of Exablate Blood-Brain Barrier Disruption for Mild Cognitive Impairment or Mild Alzheimer's Disease Undergoing Aduhelm Therapy. https://ClinicalTrials.gov/show/NCT05469009.
- Lowe SL, Willis BA, Hawdon A, et al. Donanemab (LY3002813) dose-escalation study in Alzheimer's disease. Alzheimers Dement (N Y). 2021; 7(1): e12112, doi: 10.1002/trc2.12112, indexed in Pubmed: 33614890.
- Mintun MA, Lo AC, Duggan Evans C, et al. Donanemab in Early Alzheimer's Disease. N Engl J Med. 2021; 384(18): 1691–1704, doi: 10.1056/NEJMoa2100708, indexed in Pubmed: 33720637.
- Shcherbinin S, Evans CD, Lu M, et al. Association of Amyloid Reduction After Donanemab Treatment With Tau Pathology and Clinical Outcomes: The TRAILBLAZER-ALZ Randomized Clinical Trial. JAMA Neurol. 2022; 79(10): 1015–1024, doi: 10.1001/jamaneurol.2022.2793, indexed in Pubmed: 36094645.
- Gueorguieva I, Willis BA, Chua L, et al. Donanemab Population Pharmacokinetics, Amyloid Plaque Reduction, and Safety in Participants with Alzheimer's Disease. Clin Pharmacol Ther. 2023; 113(6): 1258–1267, doi: 10.1002/cpt.2875, indexed in Pubmed: 36805552.
- Pontecorvo MJ, Lu M, Burnham SC, et al. Association of Donanemab Treatment With Exploratory Plasma Biomarkers in Early Symptomatic Alzheimer Disease: A Secondary Analysis of the TRAILBLAZER-ALZ Randomized Clinical Trial. JAMA Neurol. 2022; 79(12): 1250– -1259, doi: 10.1001/jamaneurol.2022.3392, indexed in Pubmed: 36251300.
- A Follow-On Study of Donanemab (LY3002813) With Video Assessments in Participants With Alzheimer's Disease (TRAILBLAZER-EXT). https://ClinicalTrials.gov/show/NCT04640077.
- Sims JR, Zimmer JA, Evans CD, et al. TRAILBLAZER-ALZ 2 Investigators. Donanemab in Early Symptomatic Alzheimer Disease: The TRAILBLAZER-ALZ 2 Randomized Clinical Trial. JAMA. 2023 [Epub ahead of print], doi: 10.1001/jama.2023.13239, indexed in Pubmed: 37459141.
- A Donanemab (LY3002813) Prevention Study in Participants With Alzheimer's Disease (TRAILBLAZER-ALZ 3). https://ClinicalTrials.gov/ show/NCT05026866.
- A Study of Donanemab (LY3002813) Compared With Aducanumab in Participants With Early Symptomatic Alzheimer's Disease (TRAILBLAZ-ER-ALZ 4). https://ClinicalTrials.gov/show/NCT05108922.
- A Study of Donanemab (LY3002813) in Participants With Early Symptomatic Alzheimer's Disease (TRAILBLAZER-ALZ 5). https://ClinicalTrials.gov/show/NCT05508789.
- Bateman RJ, Cummings J, Schobel S, et al. Gantenerumab: an anti-amyloid monoclonal antibody with potential disease-modifying effects in early Alzheimer's disease. Alzheimers Res Ther. 2022; 14(1): 178, doi: 10.1186/s13195-022-01110-8, indexed in Pubmed: 36447240.
- Dubois B, Feldman HH, Jacova C, et al. Research criteria for the diagnosis of Alzheimer's disease: revising the NINCDS-ADRDA cri-

teria. Lancet Neurol. 2007; 6(8): 734-746, doi: 10.1016/S1474-4422(07)70178-3, indexed in Pubmed: 17616482.

- Ostrowitzki S, Deptula D, Thurfjell L, et al. Mechanism of amyloid removal in patients with Alzheimer disease treated with gantenerumab. Arch Neurol. 2012; 69(2): 198–207, doi: 10.1001/archneurol.2011.1538, indexed in Pubmed: 21987394.
- Salloway SP, Sperling R, Fox NC, et al. Bapineuzumab 301 and 302 Clinical Trial Investigators. Two phase 3 trials of bapineuzumab in mild-to-moderate Alzheimer's disease. N Engl J Med. 2014; 370(4): 322–333, doi: 10.1056/NEJMoa1304839, indexed in Pubmed: 24450891.
- Sperling R, Salloway S, Brooks DJ, et al. Amyloid-related imaging abnormalities in patients with Alzheimer's disease treated with bapineuzumab: a retrospective analysis. Lancet Neurol. 2012; 11(3): 241–249, doi: 10.1016/S1474-4422(12)70015-7, indexed in Pubmed: 22305802.
- Delor I, Charoin JE, Gieschke R, et al. Modeling Alzheimer's Disease Progression Using Disease Onset Time and Disease Trajectory Concepts Applied to CDR-SOB Scores From ADNI. CPT Pharmacometrics Syst Pharmacol. 2013; 2(10): e78, doi: 10.1038/psp.2013.54, indexed in Pubmed: 24088949.
- A Study of Gantenerumab in Participants With Prodromal Alzheimer's Disease. https://ClinicalTrials.gov/show/NCT01224106.
- A Study of Gantenerumab in Participants With Mild Alzheimer Disease. https://ClinicalTrials.gov/show/NCT02051608.
- Klein G, Delmar P, Kerchner GA, et al. Thirty-Six-Month Amyloid Positron Emission Tomography Results Show Continued Reduction in Amyloid Burden with Subcutaneous Gantenerumab. J Prev Alzheimers Dis. 2021; 8(1): 3–6, doi: 10.14283/jpad.2020.68, indexed in Pubmed: 33336218.
- Klein G, Delmar P, Voyle N, et al. Gantenerumab reduces amyloid-β plaques in patients with prodromal to moderate Alzheimer's disease: a PET substudy interim analysis. Alzheimers Res Ther. 2019; 11(1): 101, doi: 10.1186/s13195-019-0559-z, indexed in Pubmed: 31831056.
- Roche provides update on Phase III GRADUATE programme evaluating gantenerumab in early Alzheimer's disease. 2022.
- A Study to Evaluate the Efficacy and Safety of Gantenerumab in Participants at Risk for or at the Earliest Stages of Alzheimer's Disease (AD). https://clinicaltrials.gov/show/NCT05256134.
- 43. Dominantly Inherited Alzheimer Network Trial: An Opportunity to Prevent Dementia. A Study of Potential Disease Modifying Treatments in Individuals at Risk for or With a Type of Early Onset Alzheimer's Disease Caused by a Genetic Mutation. Master Protocol DIAN-TU-001. https://clinicaltrials.gov/show/NCT01760005.
- Salloway S, Farlow M, McDade E, et al. A trial of gantenerumab or solanezumab in dominantly inherited Alzheimer's disease. Nat Med. 2021; 27(7): 1187-1196, doi: 10.1038/s41591-021-01369-8, indexed in Pubmed: 34155411.
- Doody RS, Thomas RG, Farlow M, et al. Alzheimer's Disease Cooperative Study Steering Committee, Solanezumab Study Group. Phase 3 trials of solanezumab for mild-to-moderate Alzheimer's disease. N Engl J Med. 2014; 370(4): 311-321, doi: 10.1056/NEJMoa1312889, indexed in Pubmed: 24450890.
- Logovinsky V, Satlin A, Lai R, et al. Safety and tolerability of BAN2401--a clinical study in Alzheimer's disease with a protofibril selective Aβ antibody. Alzheimers Res Ther. 2016; 8(1): 14, doi: 10.1186/s13195-016-0181-2, indexed in Pubmed: 27048170.

- Satlin A, Wang J, Logovinsky V, et al. Design of a Bayesian adaptive phase 2 proof-of-concept trial for BAN2401, a putative disease-modifying monoclonal antibody for the treatment of Alzheimer's disease. Alzheimers Dement (N Y). 2016; 2(1): 1–12, doi: 10.1016/j. trci.2016.01.001, indexed in Pubmed: 29067290.
- Wang J, Logovinsky V, Hendrix SB, et al. ADCOMS: a composite clinical outcome for prodromal Alzheimer's disease trials. J Neurol Neurosurg Psychiatry. 2016; 87(9): 993–999, doi: 10.1136/jnnp-2015-312383, indexed in Pubmed: 27010616.
- 49. Swanson CJ, Zhang Y, Dhadda S, et al. A randomized, double-blind, phase 2b proof-of-concept clinical trial in early Alzheimer's disease with lecanemab, an anti-Aβ protofibril antibody. Alzheimers Res Ther. 2021; 13(1): 80, doi: 10.1186/s13195-021-00813-8, indexed in Pubmed: 33865446.
- 50. McDade E, Cummings JL, Dhadda S, et al. Lecanemab in patients with early Alzheimer's disease: detailed results on biomarker, cognitive, and clinical effects from the randomized and open-label extension of the phase 2 proof-of-concept study. Alzheimers Res Ther. 2022; 14(1): 191, doi: 10.1186/s13195-022-01124-2, indexed in Pubmed: 36544184.
- van Dyck CH, Swanson CJ, Aisen P, et al. Lecanemab in Early Alzheimer's Disease. N Engl J Med. 2023; 388(1): 9–21, doi: 10.1056/ NEJMoa2212948, indexed in Pubmed: 36449413.
- Andrews JS, Desai U, Kirson NY, et al. Disease severity and minimal clinically important differences in clinical outcome assessments for Alzheimer's disease clinical trials. Alzheimers Dement (N Y). 2019;
 354–363, doi: 10.1016/j.trci.2019.06.005, indexed in Pubmed: 31417957.
- Lansdall CJ, McDougall F, Butler LM, et al. Establishing Clinically Meaningful Change on Outcome Assessments Frequently Used in Trials of Mild Cognitive Impairment Due to Alzheimer's Disease. J Prev Alzheimers Dis. 2023; 10(1): 9–18, doi: 10.14283/jpad.2022.102, indexed in Pubmed: 36641605.
- Thambisetty M, Howard R. Lecanemab trial in AD brings hope but requires greater clarity. Nat Rev Neurol. 2023; 19(3): 132–133, doi: 10.1038/s41582-022-00768-w, indexed in Pubmed: 36609712.
- 55. Jönsson L, Wimo A, Handels R, et al. The affordability of lecanemab, an amyloid-targeting therapy for Alzheimer's disease: an EADC-EC viewpoint. Lancet Reg Health Eur. 2023; 29: 100657, doi: 10.1016/j. lanepe.2023.100657, indexed in Pubmed: 37251789.
- Cummings J, Apostolova L, Rabinovici GD, et al. Lecanemab: Appropriate Use Recommendations. J Prev Alzheimers Dis. 2023; 10(3): 362– 377, doi: 10.14283/jpad.2023.30, indexed in Pubmed: 37357276.
- Wach-Klink A, Iżycka-Świeszewska E, Kozera G, et al. Cerebral microbleeds in neurological practice: concepts, diagnostics and clinical aspects. Neurol Neurochir Pol. 2021; 55(5): 450–461, doi: 10.5603/ PJNNS.a2021.0058, indexed in Pubmed: 34379320.
- Jönsson L, Wimo A, Handels R, et al. The affordability of lecanemab, an amyloid-targeting therapy for Alzheimer's disease: an EADC-EC viewpoint. Lancet Reg Health Eur. 2023; 29: 100657, doi: 10.1016/j. lanepe.2023.100657, indexed in Pubmed: 37251789.
- Piller C. Report on trial death stokes Alzheimer's drug fears. Science. 2023; 380(6641): 122–123, doi: 10.1126/science.adi2242, indexed in Pubmed: 37053319.
- 60. AHEAD 3-45 Study: A Study to Evaluate Efficacy and Safety of Treatment With Lecanemab in Participants With Preclinical Alzheimer's Disease and Elevated Amyloid and Also in Participants With Early Preclinical Alzheimer's Disease and Intermediate Amyloid. https:// clinicaltrials.gov/show/NCT04468659.

- A Study of LY3372993 in Participants With Alzheimer's Disease (AD) and Healthy Participants. https://clinicaltrials.gov/show/ NCT04451408.
- Jin Y. Safety and amyloid plaque reduction effects of remternetug in patients with Alzheimer's disease: interim analysis from a phase 1 study. AD/PD. Gothenburg, Sweden 2023.
- A Study of Remternetug (LY3372993) in Participants With Alzheimer's Disease (TRAILRUNNER-ALZ 1). https://clinicaltrials.gov/show/ NCT05463731.
- Roberts M, Sevastou I, Imaizumi Y, et al. Pre-clinical characterisation of E2814, a high-affinity antibody targeting the microtubule-binding repeat domain of tau for passive immunotherapy in Alzheimer's disease. Acta Neuropathol Commun. 2020; 8(1): 13, doi: 10.1186/ s40478-020-0884-2, indexed in Pubmed: 32019610.
- A Study to Assess Safety, Tolerability, Pharmacokinetics (PK), Immunogenicity, and Pharmacodynamics (PD) of Intravenous Infusions of E2814 in Healthy Participants. https://clinicaltrials.gov/show/ NCT04231513.
- Ji C, Sigurdsson EM. Current Status of Clinical Trials on Tau Immunotherapies. Drugs. 2021; 81(10): 1135–1152, doi: 10.1007/s40265-021-01546-6, indexed in Pubmed: 34101156.
- Dominantly Inherited Alzheimer Network Trial: An Opportunity to Prevent Dementia. A Study of Potential Disease Modifying Treatments in Individuals With a Type of Early Onset Alzheimer's Disease Caused by a Genetic Mutation (DIAN-TU). https://clinicaltrials.gov/show/ NCT05269394.
- Salloway SP, Sevingy J, Budur K, et al. Advancing combination therapy for Alzheimer's disease. Alzheimers Dement (N Y). 2020; 6(1): e12073, doi: 10.1002/trc2.12073, indexed in Pubmed: 33043108.
- Middleton G, Silcocks P, Cox T, et al. Gemcitabine and capecitabine with or without telomerase peptide vaccine GV1001 in patients with locally advanced or metastatic pancreatic cancer (TeloVac): an open-label, randomised, phase 3 trial. Lancet Oncol. 2014; 15(8): 829-840, doi: 10.1016/S1470-2045(14)70236-0, indexed in Pubmed: 24954781.
- Park HH, Lee KY, Kim S, et al. Novel vaccine peptide GV1001 effectively blocks β-amyloid toxicity by mimicking the extra-telomeric functions of human telomerase reverse transcriptase. Neurobiol Aging. 2014; 35(6): 1255–1274, doi: 10.1016/j.neurobiolaging.2013.12.015, indexed in Pubmed: 24439482.
- Park HH, Yu HJ, Kim S, et al. Neural stem cells injured by oxidative stress can be rejuvenated by GV1001, a novel peptide, through scavenging free radicals and enhancing survival signals. Neurotoxicology. 2016; 55: 131–141, doi: 10.1016/j.neuro.2016.05.022, indexed in Pubmed: 27265016.
- 72. Koh SH, Kwon HS, Choi SH, et al. Efficacy and safety of GV1001 in patients with moderate-to-severe Alzheimer's disease already receiving donepezil: a phase 2 randomized, double-blind, placebo-controlled, multicenter clinical trial. Alzheimers Res Ther. 2021; 13(1): 66, doi: 10.1186/s13195-021-00803-w, indexed in Pubmed: 33771205.
- GV1001 Subcutaneous(SC) for the Treatment of Mild to Moderate Alzheimer's Disease (AD). https://clinicaltrials.gov/show/ NCT05189210.
- GV1001 Subcutaneous for the Treatment of Moderate to Severe Alzheimer's Disease (AD). https://clinicaltrials.gov/show/ NCT05303701.
- 75. Bednarz K, Siuda J. Alzheimer's disease and type 2 diabetes mellitus: similarities in pathomechanisms lead to therapeutic opportunities.

Neurol Neurochir Pol. 2021; 55(5): 418-428, doi: 10.5603/PJNNS. a2021.0056, indexed in Pubmed: 34355790.

- During MJ, Cao L, Zuzga DS, et al. Glucagon-like peptide-1 receptor is involved in learning and neuroprotection. Nat Med. 2003; 9(9): 1173–1179, doi: 10.1038/nm919, indexed in Pubmed: 12925848.
- Zhang ZQ, Hölscher C. Novel dual GLP-1/GIP receptor agonists show neuroprotective effects in Alzheimer's and Parkinson's disease models. Neuropharmacology. 2018; 136(Pt B): 251–259, doi: 10.1016/j. neuropharm.2018.01.040, indexed in Pubmed: 29402504.
- Salameh TS, Rhea EM, Talbot K, et al. Brain uptake pharmacokinetics of incretin receptor agonists showing promise as Alzheimer's and Parkinson's disease therapeutics. Biochem Pharmacol. 2020; 180: 114187, doi: 10.1016/j.bcp.2020.114187, indexed in Pubmed: 32755557.
- 79. A Research Study Investigating Semaglutide in People With Early Alzheimer's Disease (EVOKE). https://clinicaltrials.gov/show/NCT04777396.
- A Research Study Investigating Semaglutide in People With Early Alzheimer's Disease (EVOKE Plus). https://clinicaltrials.gov/show/ NCT04777409.